Author's response to reviews

Title: What factors explain the number of physical therapy treatment sessions in patients referred with low back pain; a multilevel analysis

Authors:
Ilse CS Swinkels (i.swinkels@nivel.nl)
Raymond H Wimmers (r.wimmers@hartstichting.nl)
Peter P Groenewegen (p.groenewegen@nivel.nl)
Wil JH van den Bosch (w.vandenbosch@hag.umcn.nl)
Joost Dekker (j.dekker@vumc.nl)
Cornelia HM van den Ende (e.vandenende@cukz.umcn.nl)

Version: 4  Date: 11 November 2005

Author's response to reviews:

Dear editor,

Thank you for approving my manuscript (MS 1106594052685272) and the thorough review process.

According to your instructions I have checked whether the title, abstract and authors' names in the manuscript are identical to those in the submission system. Furthermore I have removed the keywords section. I have renamed Box 1 and Box 2 into Table 1 and Table 2 and I have renumbered the other tables. References to these tables in the text are renumbered as well. The figure and table placement text are removed.

With kind regards,

Ilse Swinkels